Author's response to reviews

Title: Large-scale, prospective, observational studies in patients with psoriasis and psoriatic arthritis: A systematic and critical review.

Authors:

Sue Langham (suelangham@phmrconsulting.com)
Julia Langham (julia_langham@hotmail.com)
Hans-Peter Goertz (hans-peter.goertz@novartis.com)
Mark Ratcliffe (markratcliffe@phmrconsulting.com)

Version: 4 Date: 18 January 2011

Author's response to reviews:

We provided the clarification of using 100 cut-off in the discussion section (when discussing weaknesses) to show that only one relevant paper was excluded based on this criteria. This has also now been provided in the methods section in the attached updated manuscript – all figures remain the same.